\documentclass[11pt]{article}
\usepackage{linalg}
\usepackage[pdftex]{graphicx}
\parindent=0in
\parskip=1ex
\begin{document}

\begin{center}\large\bf
Linear Algebra 2270-2 \\
\normalsize\rm
Due in Week 6
\end{center}

The sixth week continues the work from chapter 2 and starts chapter 3. Here's the list of problems, all of which have book answers.

\begin{description}

\item[{\bf Section 2.4}.] Exercises 3, 5, 15, 21, 23

\item[{\bf Section 2.5}.] Exercises 3, 7, 11, 25

\item[{\bf Section 3.1}.] Exercises 5, 9, 11, 15, 25, 27, 29, 31

\end{description}

\end{document}
